www.cambridge.org/pab

### Article

**Cite this article:** Syverson, V. J.P., and D. Prothero (2025). Reevaluating climate change responses in Rancho La Brea birds and mammals: new dates and new data. *Paleobiology*, 1–14. https://doi.org/10.1017/pab.2024.37

Received: 31 January 2024 Revised: 05 August 2024 Accepted: 16 August 2024

Handling Editor: Margaret Yacobucci

Corresponding author: V. J. P. Syverson; Email: vsyverson@gmail.com

© The Author(s), 2025. Published by Cambridge University Press on behalf of Paleontological Society. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http:// creativecommons.org/licenses/by/4.0), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.



# Reevaluating climate change responses in Rancho La Brea birds and mammals: new dates and new data

## Val J. P. Syverson<sup>1</sup> 💿 and Donald Prothero<sup>2</sup>

<sup>1</sup>Department of Life and Environmental Sciences, University of California, Merced, Merced, California 95343, U.S.A.
<sup>2</sup>Department of Geological Sciences, California State Polytechnic University, Pomona, Pomona, California 91768, U.S.A.

### Abstract

The Pleistocene/Holocene transition furnishes a classic example of apparent evolutionary stasis during a period of major environmental change, where observed environmental clines might predict evolutionary change. We have previously attempted to assess whether or not body size and shape were static in the extensive Quaternary vertebrate fauna of Rancho La Brea (RLB). However, the validity of time-series studies depends on dating, and there are indications that previous approaches based on pit mean radiocarbon ages may be misleading. Here we have compiled and recalibrated all available RLB radiocarbon ages, reanalyzed our morphometric data using a novel method for bootstrap resampling of calibrated C age distributions to estimate a time series of populations, and fit the time series with a range of simple evolutionary time-series models.

Although the shortness of our time series tends to favor nondirectional models and is insufficient to allow reliable discrimination between punctuated and gradual change, the results can still be clearly interpreted. The population means for most anatomical elements in most species at RLB do genuinely appear to be static through the Pleistocene/Holocene transition, as previously published. Some species exhibit previously undetected changes in population mean size and shape, including *Smilodon, Gymnogyps*, and *Equus*. However, the timing of change is variable among the non-static species and generally does not correspond to changes in temperature, and thus resists a Bergmann's rule interpretation. Considering the species by ecological category may reveal more about the effects of climate regime shifts.

### **Non-technical Summary**

At the end of the last ice age, there was a major worldwide shift in climate, and animals may have had to adapt to the changes in order to survive. Many modern animals are larger and stockier in parts of their range with colder climates, so we might expect Ice Age animals to be larger and stockier than their relatives living in a warmer, more modern climate. The fossils at Rancho La Brea formed during this time, so we can test this hypothesis by comparing older La Brea fossils to newer ones. However, this requires accurate estimates of their ages, which can be complicated, because each tar pit contains a mixture of older and younger fossils. We have already taken measurements on a large number of fossils from many species. In this paper, we test a new method for estimating population average sizes over time, based on all the radiocarbon dates from each pit.

Our results show that most of the birds and mammals at Rancho La Brea did not change much as the climate warmed, and those that did change size or shape did not all get smaller or change size at the same time that the climate was changing. Using our new method produced some new results that we missed the first time, but that other authors had noticed.

### Introduction

### Punctuated Equilibrium, Stasis, and the Last Glacial Maximum (LGM)

One of Darwin's fundamental insights, gained primarily through observations of modern populations, is that the frequencies of traits in populations change in response to their environments. How quickly and when they respond, though, is a question best answered with the fossil record. One of the major shifts in paleontological thinking about the tempo and mode of evolution, initiated 50 years ago, forms the topic of this issue (Eldredge and Gould 1972; Gould and Eldredge 1993). The key to the punctuated equilibrium model of evolution is the observation that evolutionary change in populations is relatively rare, even during intervals of environmental change, and occurs mainly during speciation. This offered a possible solution to the apparent "paradox of stasis," in which evolution appeared to be rapid and common in living populations, but much slower and rarer on paleontological timescales (e.g., Estes and Arnold 2007). Indeed,



the Pleistocene glacial–interglacial cycles furnish one of the first examples of paleontological stasis noted in print, shortly after the publication of *On the Origin of Species*: "If we cast a glance back on the long vista of physical changes which our planet has undergone since the Neozoic Epoch, we can nowhere detect signs of a revolution more sudden and pronounced, and more important in its results, than the intercalation and subsequent disappearance of the Glacial period. Yet the mammoth lived before it, and passed through the ordeal of all the hard extremities which it involved, bearing his organs of locomotion and digestion all but unchanged" (Falconer 1863: p. 44). In creating a theoretical framework for interpreting stasis, however, Eldredge and Gould invited paleontologists to investigate it explicitly and interpret their results in terms that explored the concept of tempo variation and timescale dependency in evolution.

Paleontologists' discussion of variations in the tempo and mode of evolution has by now transcended simple comparison of the relative frequencies of gradual versus punctuated change; modern studies are more likely to address the temporal scaling of evolutionary change in general, using a variety of new analytical techniques. As predicted by Stanley (1975), Vrba and Gould (1986), Gould (2002), and Jablonski (2008), species rarely show gradual directional change over million-year timescales. Although ephemeral evolutionary change is common and uncorrelated with time on sub-million-year timescales, phenotypic changes rarely persist and accumulate for much longer than that (Estes and Arnold 2007; Uyeda et al. 2011). This discontinuity between short and long timescales has been explained at a population genetics level by the failure of most adaptations to sweep through populations before the local optimum shifts away from them again, unless reproductive isolation allows them to become fixed in a new species. This pattern is termed "ephemeral divergence" by Futuyma (2010), who suggests that it may account for the observed association between climatic fluctuation and morphological stasis.

Ecological niche occupation traits appear to be at least as conservative as morphological traits; documented examples of niche shift are exceedingly rare, even during speciation events. Results from ecological niche model studies indicate that environmental changes on sub-million-year timescales usually result only in range shifts, even in lineages that split by environmentally driven vicariant or allopatric speciation. Climatic niches are highly phylogenetically conserved: little niche shift occurs on the 10<sup>3</sup>–10<sup>4</sup> year timescale, either during speciation events (Wiens 2004; Peterson 2011; Pyron et al. 2015) or after several million years of separation between sister species, with significant niche shift occurring only after an order of magnitude longer (Peterson et al. 1999). Even in a case where genetic data indicate intraspecific evolution with a large subspecies being replaced by a small one during the Pleistocene/Holocene transition, ecological niche modeling indicates that climate was probably not the cause (Prost et al. 2013).

The apparent paradox of the lack of change on the sub-millionyear timescales, but the many examples of change on the multimillion-year timescale, can best be explained by species sorting in both clades and assemblages (Prothero et al. 2012). Studies that demonstrate apparent change in species with climate over multimillion-year timescales (Janis 1989, 1993; Stucky 1992; Prothero 1999) treat species as discrete units with static properties and sum the species diversity (treated as discrete units with their own properties) in bins of 2–3 Myr in duration or longer, so they are unable to see the fine-scale stasis observed on sub-million-year timescales. The apparent climatic response of species in these multi-million-year studies is not direct morphological change tracking climate on short timescales, but the increase and decrease in number of species in each time bin.

In the light of these insights about the time-scaling of evolutionary change, widespread morphological stasis in vertebrates during the  $\sim 10^5$  years of the Pleistocene/Holocene transition is unsurprising, and ecological stasis is exactly what would be expected. Studies in a variety of taxa back this prediction, including mammals (e.g., Barnosky 2005); birds (e.g., Zink et al. 2004); plants (e.g., Willis et al. 2004); and reptiles, amphibians, and fishes (Avise et al. 1998). The results from climatic niche–based species distribution models instead predict that the changes in climatic suitability since the last interglacial (LIG, ~120 ka) should be accommodated by range shifts (Graham et al. 1996), with most extirpated species being replaced by ecologically similar relatives (McGill et al. 2005).

### Tempo and Mode at Rancho La Brea (RLB)

The fossils from the famous tar pit Lagerstätte at RLB provide an opportunity to study the pace of adaptation to environmental change on the 10–100 kyr timescale in an unusually rich and well-preserved Late Pleistocene to Holocene fossil vertebrate fauna. These deposits formed in an asphalt seep in the Los Angeles Basin in Southern California over the last ~50 kyr (Fox et al. 2023), which spans the last full interglacial–glacial cycle of the Pleistocene and the entirety of the Holocene, and preserve a wide range of vertebrates, insects, and plants. The regional vegetation reflects the Pleistocene/Holocene climate transition, shifting from oak–chaparral at the LIG to juniper–pine forest at the LGM and back to oak–chaparral in the Holocene (Heusser 1998; Heusser et al. 2015). Thus, RLB provides a detailed record of exactly the sort of submillion-year climatic variability that is thought to produce ephemeral evolutionary change in morphological traits.

Ecological niche conservatism has been specifically documented in the vertebrate fauna of RLB. Ecological niche models of extant bird species found as fossils at RLB indicate that their environmental preferences and life histories have changed minimally since the LGM: niche models based on modern occurrence data predicted the presence at 21 ka of 95% of the species within 100 km of RLB (Zink et al. 2020). Likewise, dietary isotope data indicate that the diets of RLB fossil individuals of extant mammal species from the LIG are similar to those of their modern conspecifics; *Canis latrans*, with its unusual dietary flexibility, is the exception that proves the rule (Fuller et al. 2020). Obviously, neither of these studies was suitable for examination of extinct species, but both indicate that surviving species inhabit a similar niche to their end-Pleistocene relatives.

The authors of this paper and our collaborators have published 22 separate papers on individual species or small groups of similar species (Table 1) based on the radiocarbon ages and paleoclimate proxies that were available when we began the project. In 2012, we reviewed the results from the first 12 species we studied (Prothero et al. 2012), finding an overwhelming signal of morphological stasis. Since then, we have measured elements of 23 more species, mostly birds, all with similar results. Over the same period, however, many more RLB radiocarbon dates and regional paleoclimate datasets have been published, as well as a convincing argument against the validity of using pit mean ages (Fuller et al. 2014). In the present study, we use a newly revised radiocarbon chronology for RLB, a newly revised paleoclimate interpretation for the Los Angeles Basin, and our previously published morphometric datasets to assess whether size or robustness in the vertebrate fossils of RLB

**Table 1.** Species previously published on as part of this series of studies, in order of publication date. Daggers (†) indicate species represented in study by fossil specimens from La Brea Tar Pits Museum; asterisks (\*) indicate species in study represented by modern specimens from other collections.

Species	Reference
†Gymnogyps amplus, *Gymnogyps californianus	Syverson and Prothero (2010)
†Equus occidentalis, †Camelops hesternus	DeSantis and Prothero (2011)
†Haliaeetus leucocephalus, †Meleagris californica, †Caracara prelutosa (= Polyborus prelutosus = Caracara cheriway at Rancho La Brea; see discussions of taxonomy in Fuchs et al. 2012; Oswald et al. 2019)	Fragomeni and Prothero (2011)
†Panthera atrox, †Smilodon fatalis	Madan et al. (2011)
†*Aquila chrysaetos	Molina and Prothero (2011)
†Paramylodon harlani	Prothero and Raymond (2011)
†Bison antiquus	Raymond and Prothero (2011)
†Aenocyon dirus	Prothero et al. (2012)
†*Bubo virginianus	Madan et al. (2015)
†Teratornis merriami	Gillespy et al. (2016)
<i>†Coragyps occidentalis, *Coragyps atratus</i>	Long et al. (2016)
†*Tyto alba	Madan et al. (2016)
†*Asio otus, †*Athene cunicularia	Madan et al. (2019)
<pre>†*Felis concolor, †*Lynx rufus</pre>	Balassa et al. (2020)
†*Corvus corax, †*Pica nuttalli, †*Sturnella neglecta	Long et al. (2020)
†*Buteo jamaicensis	Balassa et al. (2022)
†*Buteo swainsoni	Cleaveland et al. (2022)
†Spizaetus grinnelli	DeAnda et al. (2022)
†Circus hudsonius	Marriott et al. (2022)
†Neophrontops americanus, *Neophron percnopterus	Olson et al. (2022)
†Buteogallus fragilis	Santos et al. (2022)
†*Buteo regalis	Watmore and Prothero (2022)

can be correlated with the changes in regional climate during the last glacial–interglacial interval.

The purpose of the present study is a review and extended correction to this series of papers, in which we reevaluate all parts of the preceding studies. The hypothesis and underlying theory are reevaluated in this section; the suitability and quality of the data are discussed in the next section, along with a reconsideration of the analytical methods and statistical framework. The results of the analysis and our conclusions, therefore, also inevitably demanded a reassessment. The ensuing sections provide a general interpretation of the contribution that we hope to make to the discussion of tempo and mode in evolution by this intensive focus on the relationship between climate and vertebrate body size in the RLB paleoecosystem.

### Hypotheses and Predictions

Any study based on a single locality can necessarily only detect local trends. Thus, we may potentially distinguish change from stasis in the local population with this dataset, but we cannot tell range shifts among populations apart from species-level evolutionary change. As such, we must stick to the question of whether the bird and mammal populations at RLB exhibited any morphological response to the climate transition; we cannot claim to evaluate whether that morphological response, be it stasis or change, is a biogeographic or an evolutionary one. Based on the studies already highlighted, we predict that we will not see significant morphological change in most populations.

In recent meta-analyses of evolutionary mode, directional evolution was most strongly supported for between 9% (Hunt et al. 2015) and 14% (Voje 2016) of fossil time series. Based on this result, if all the time series were independent, we would expect to find directional evolution in between 40 and 63 of our 455 morphometric measurements (Supplementary Table A). However, our data are likely to be autocorrelated, because they have a great deal of structure: these 455 time series were measured from 57 skeletal elements across 30 species, all from the same set of 26 pit deposits. Comparison to a numeric estimate based on compilations of independent fossil time series from various taxa and intervals is therefore probably not useful. Instead, we should be looking for clusters of directional evolutionary trends that follow the structure of the data. For example, we might expect to find multiple bones with similar trends in a single species; ecologically similar species might all remain stable or change together.

Modern zoology supplies us with a prediction for the timing and direction of any changes we might find, based on known climaterelated body-size and body-shape clines in extant populations of a number of RLB species and their close relatives. One of the most commonly recognized of these, Bergmann's rule, describes a pattern of larger and stockier individuals in cold-adapted populations and smaller, longer-limbed individuals in warmer regions (Mayr 1956). The list of species in which the effect of Bergmann's rule has been observed includes various extant ungulates (Nowak 1999), felids (Sunquist and Sunquist 2002), birds of prey (Johnsgard 1990), and passerines (Gill 2003; Fan et al. 2019), as well as some fossil species such as the Paleogene equid *Sifrhippus* (Secord et al. 2012) and the Late Pleistocene to Holocene woodrat Neotoma cinerea (Smith et al. 1995). Several separate studies have estimated that Bergmann's rule applies to around 70% of living mammal species, more strongly at higher latitudes (Ashton et al. 2000; Rodríguez et al. 2008). However, carnivorans in general may not follow it (Yom-Tov and Geffen 2011). Indeed, two geometric morphometric studies of RLB carnivorans did not recover this pattern across time: postcranial elements in *Canis latrans* decrease in size in apparent response to the megafaunal extinction, not to deglaciation (Meachen and Samuels 2012), and mandible size in Smilodon fatalis at RLB actually inverts the pattern, with smaller ancestral forms in cold intervals and larger derived forms in warmer times (Meachen et al. 2014). Still, Bergmann's rule furnishes a reasonable starting point for what overall direction and timing of change, if any, we might expect to see in our RLB vertebrate morphological data.

### **Data and Methods**

No original data were collected for this study. Instead, we used a novel method to assign ages to already-measured specimens, then modeled the evolutionary dynamics of the resulting time series. The overall procedure for this study was as follows: (1) Morphometric data for various RLB vertebrates were compiled from the results of our 22 previous studies. (2) We compiled a database of RLB radiocarbon dates and used it to estimate an age probability distribution for each pit. (3) Each specimen was then assigned an age drawn at random from the appropriate pit age distribution; this step was repeated 1000 times. (4) Time bins were determined based on a variety of regional climate reconstructions. The mean and variance of each measured parameter was then calculated for each time bin. (5) From these, we constructed a time series for each measurement. (6) Finally, a set of evolutionary models was tested against each time series. The following subsections describe each of these steps in detail.

### Morphometric Data

Linear measurements of skeletal elements for a variety of the most common vertebrate species found in the RLB asphalt deposits were taken over the course of 12 years by 13 different investigators, using digital or dial calipers. The species and bones measured, counts, and original publications are listed in Supplementary Table A. These include 17 bird and 9 mammal species, of which 14 are extant and 12 extinct. All skeletal elements with 10 or more specimens available were measured. For 10 species, multiple elements were available, but for the majority of bird species this meant only the tarsometatarsus (TMT). The number of individual bones measured from each species ranges from 21 (Pica nuttalli) to 1496 (Smilodon fatalis), and the number of specimens of each skeletal element ranges from 10 (tooth M1 of Puma concolor) to 373 (astragalus of S. fatalis). The measurements taken on each element were chosen to best capture variability in body size and are documented in detail in the original papers (Table 1). For this study, to facilitate comparisons between elements and species, we also log-transformed the data and then did a principal components analysis. The signs of the principal component (PC) rotation matrices were flipped to orient them to the measurements so that the positive direction of each PC corresponded to the positive direction of the measurements. The variation captured in the first two PCs varies from 50% to 100% (scree box plot: Supplementary Fig. B).

The question addressed in this series of studies has been whether climate changes have affected body size or shape in the RLB population of each species. A number of the papers have used measurements of modern conspecifics from zoological collections for comparison. However, the original authors measured recent specimens collected from a variety of different locations and held in a number of different repositories, and many did not record the information necessary to determine whether they were collected in Southern California. The differences between the modern climates of various collection areas very well may eclipse those between glacial and interglacial RLB, so this risks obscuring the signal of any potential population-level effect of climate on morphology. In the present study, we therefore omit modern specimens from all quantitative analyses.

### Radiocarbon Estimation of Pit Ages

Our previous studies used the RLB<sup>14</sup>C pit ages reported by Marcus and Berger (1984) and O'Keefe et al. (2009), later adding those compiled by Fuller et al. (2014, 2015). We then assigned the age of all specimens within a pit to the mean age of that pit. The ages used in these previous studies are given in Table 2. However, as discussed by Fuller et al. (2014), this is not good statistical practice, owing to the unusual taphonomic characteristics of the RLB pit deposits. Each "pit" is a single seep of asphalt, a few meters wide and a few meters deep. Within each pit, severe taphonomic time averaging has taken place: vertical relationships between the specimens in a pit have no significance for superposition, and therefore no stratigraphic sequence can be established (Woodard and Marcus 1973; Friscia et al. 2008; O'Keefe et al. 2009). Each pit must therefore be treated as a single time-averaged deposit. Furthermore, each pit may document multiple disjoint intervals of deposition with irregularly varying intensity, depending on factors such as asphalt flow and surface temperature (Friscia et al. 2008; Fuller et al. 2014). The mean age of the specimens in a pit therefore may bear little relation to the age of any individual specimen (Fuller et al. 2014), which makes it an unsuitable estimator for measurements taken from individual specimens.

For this study, the pit chronology was revised based on the <sup>14</sup>C dates in the studies already mentioned (Marcus and Berger 1984;

**Table 2.** Median and boundary (95% CI) radiocarbon ages for pits at Rancho La Brea, calculated using OxCal and expressed in calendar years BP, along with the age values used in the previous studies listed in Table 1. These pits are those for which we have more than 10 measured specimens and at least one high-quality radiocarbon date on a vertebrate sample. For full list of <sup>14</sup>C dates used, see Supplementary Table B.

Pit no.	No. <sup>14</sup> C dates used	Cal <sup>14</sup> C max. age	Cal <sup>14</sup> C median age	Cal <sup>14</sup> C min. age	Ages (ka) used in previous papers
3	41	23,640	15,550	13,845	18; 22
4	13	39,660	26,680	12,970	14; 16
9	0	_	—	—	13
10	9	19,860	11,990	4,245	9
13	4	18,815	18,485	18,160	16
16	8	43,580	28,520	13,165	27; not used
60	6	34,220	23,305	12,770	21
61/67	175	15,105	13,075	11,040	11
77	4	41,955	25,690	9,290	35
81	1	44,587	19,780	10,187	17
91	78	44,750	31,210	85	34
98–99	0	_	_	_	29

O'Keefe et al. 2009; Fuller et al. 2014, 2015), as well as in several others published more recently (Holden and Southon 2016; Holden et al. 2017; Fuller et al. 2020; Fox et al. 2023; O'Keefe et al. 2023). The ages of specimens were then estimated from the ages of all the specimens in the pit, based on a bootstrapping procedure detailed in the "Age Resampling and Time-Series Computation" section. All analyses were performed using R statistical software (v. 4.2.3). The resulting revised age ranges are given alongside the previously used pit mean age values in Table 2.

*Revised Pit Age Estimates.* A list of radiocarbon dates was assembled from the compilation by Fuller et al. (2015) and from the lists published in five subsequent studies (Holden and Southon 2016; Holden et al. 2017; Fuller et al. 2020; Fox et al. 2023; O'Keefe et al. 2023), provided in Supplementary Table B. After omitting the single-sided dates and two ages from Ward et al. (2005) that were older than the maximum range of the IntCal20 calibration curve, we calibrated the remaining dates in OxCal 4.4 (Bronk Ramsey 2001) using R package oxcAAR (Hinz et al. 2021) based on the IntCal20 calibration curve (Reimer et al. 2020). These dates are depicted in Figure 1, sorted by pit. All dates originally published as Pit 61, 67, or 61/67 have been combined into Pit 61/67; the radiocarbon date distributions for these three categories are identical.

For the purposes of estimating pit age distributions, we omitted dates with reliability rated below 2 by Fuller et al. (2015), dates on the >30 kDa fraction from Fuller et al. (2014), and dates from pits for which we have no measured specimens. Vertebrates are the focus of this study, so we also chose to omit dates from plant (cellulose) and insect (chitin) material and use only dates from vertebrate material

(bone and bone collagen). The chitin ages are younger and wider ranging than those from associated bone, which Holden et al. (2017) suggests may reflect differences in the taphonomic processes operating on insects and large vertebrates: small organisms may become entrapped in asphalt pools too shallow to immobilize larger ones. Additionally, decomposers may lay eggs in vertebrate remains that resurface long after their death due to the churning of the asphalt. The wood and wood charcoal ages are less obviously different from the bone ages, but in some pits, the wood dates significantly expand the time range of the pit, and taphonomic differences between wood and other remains are known to produce systematic offsets (the "old-wood problem"; Schiffer 1986).

After filtering out the categories noted, we were left with 339<sup>14</sup>C dates usable for our analysis (black dots in Fig. 1); pits 2 and 9 were omitted, having no <sup>14</sup>C ages on vertebrate material. These dates were used to estimate an age distribution for each pit using OxCal 4.4 (Bronk Ramsey 2001) with custom scripts (see Supplementary Appendix) and the R package oxcAAR (Hinz et al. 2021). The age distributions were calculated as Bayesian kernel density estimates (KDEs) of all the calibrated dates from that pit, following the advice of Carleton and Groucutt (2021). For pits 36 and 81, each with only a single date available, the posterior age distribution of that date was used instead. These age distributions are plotted together (normalized) in Figure 2 and separately without normalization in Supplementary Figure A. The pit KDEs were used as the basis for estimating the age distribution of the specimens, as detailed in the "Age Resampling and Time-Series Computation" section.



Figure 1. Distribution of published Rancho La Brea (RLB) radiocarbon ages by pit. Only dates in black were used to generate the bootstrapped time series used in the subsequent analysis; dates in gray, blue, and green were excluded (see main text for reasons). Vertical lines indicate climate bins used in this analysis; solid lines are the primary boundaries, dotted lines are the subdivided bins. Red vertical lines indicate the last glacial maximum (LGM) depositional hiatus postulated by Fuller et al. (2014).



# Pit age probability distributions (normalized)

Figure 2. All pit age probability distributions based on dates in Fig. 1. Distributions have been normalized to a maximum value of 1 and overlaid to illustrate timing. Heavier vertical dashed lines correspond to time boundaries between the five main climate bins; lighter ones correspond to boundaries between the nine subdivided climate bins.

### Paleoclimate Reconstruction

In all our previous studies, we based our reconstruction of RLB paleoclimate on those that were available at the time (Heusser 1998; Coltrain et al. 2004; Bochenski and Campbell 2006). For this study, we have instead constructed a local climate timeline based on a combination of recent papers, and we used this to designate transition points between climate intervals. A number of late Pleistocene climate estimates have been published for Southern California, Lake Elsinore, and RLB, based on a variety of proxies (Heusser 1995, 1998; Heusser and Sirocko 1997; Ward et al. 2005; Kirby et al. 2013, 2018; Fuller et al. 2014; Heusser et al. 2015; Holden et al. 2017; Fox et al. 2023; O'Keefe et al. 2023). Here, we have used these to estimate major climate phases at RLB in order to choose breaks between time bins for our time-series analysis that best correspond to the intervals of environmental change. The break points given in these publications are shown in Figure 3.

The multi-proxy climate reconstruction of the interval from 15.6 ka to 10 ka by O'Keefe et al. (2023) was generated using break-point analysis of multiple climate time series to infer local climate changes. We have not attempted such a rigorous reconstruction here, although we suspect that the data available may be sufficient for such an analysis. Instead, we used the break points described in the text of the original publications to choose our climate intervals by eye, weighting proxies by their proximity to the Los Angeles Basin. After testing revealed that our original five climate bins produced time series that were too short for a model testing approach, as discussed in the "Results" section, we also defined a set of nine bins. The break points transcribed from the literature and our resulting climate intervals are given in Supplementary Table C and depicted in the bottom two lines of Figure 3. The vertical dashed lines in Figures 1 and 3 give the boundaries between the main five climate intervals.

Our interval boundaries were chosen to capture the consensus in the regional climate reconstruction literature. Interval 1 ("MIS 3," 50–29 ka) corresponds to the cooling from the LIG into the LGM; lacking any distinct regional climatic break points, we simply divided it in half at 39.5 ka. Interval 2 ("LGM," 29-20 ka) goes up to the end of the cold and dry period according to Heusser et al. (2015). We split it in half at 25.5 ka, such that LGM-A contains the 27.5-25.5 ka drought noted from the Lake Elsinore cores (Heusser et al. 2015; Feakins et al. 2019) and LGM-B contains most of the 22.5–19.3 ka RLB depositional hiatus (18–16.5 uncalibrated <sup>14</sup>C ka BP) (Fuller et al. 2014). Interval 3 ("Deglacial") covers the cold and wet period from 20 to 15 ka; the midpoint split at 17.5 ka is again simply for convenience. Interval 4 ("Transitional," 15-11 ka) corresponds to the interval analyzed in detail by O'Keefe et al. (2023), with the first half (15-13.3 ka) approximately coinciding with the warm Bølling-Allerød interstadial, and the second (13.3-11 ka) to the wildfire interval of O'Keefe et al. (2023) and the Younger Dryas. The remainder of the time (11–4 ka) is a single Holocene interval, as there were not enough vertebrate dates available to justify splitting it.

### Age Resampling and Time-Series Computation

Due to the severely nonparametric distribution of ages in each pit, as discussed earlier, we used a bootstrapping procedure to estimate the age distributions of the measurements. This procedure makes the following assumptions: (1) complete time averaging within pits; (2) unbiased sampling of specimens for dating; (3) reasonably complete sampling of each pit; and (4) a single age distribution that applies equally to all taxa.

For each specimen (individual measured bone) in the database, an age was drawn at random from the computed probability distribution for its pit and assigned to that specimen; this procedure was replicated 1000 times. This bootstrapped dataset was used to generate averaged time series of each measurement on each element, as follows: For each time bin in the time series, the average age was computed as the grand mean of assigned specimen ages in that time bin among all bootstrap replicates, and the average sample size as the mean number of specimens in the time bin among all replicates. Then, again for each time bin in the time series, the



Figure 3. Timeline of regional climate events and local climate chronology based on Heusser 1995, 1998; Heusser and Sirocko 1997; Ward et al. 2005; Kirby et al. 2013, 2019; Fuller et al. 2014; Heusser et al. 2015; Holden et al. 2017; Feakins et al. 2019; Fox et al. 2023; O'Keefe et al. 2023. The numeric boundary values used to generate this figure are given in Supplementary Table C. DG, Deglacial; Hol., Holocene; HS-1, Heinrich Stadial 1; LGM, last glacial maximum; MIS, Marine Isotope Stage; RLB, Rancho La Brea; Tr., Transitional.

average for each measurement taken on that element was computed as the grand mean of the values of that measurement for each replicate, and the average variance was computed as the mean of the variances for each replicate. These time series were generated for all direct measurements as well as for the first and second PCs of the variation for each specimen in each species. To assess the possible bias toward stasis introduced by this averaging procedure, 250 individual replicate time series for each element were also subjected to model fitting (see next section).

### **Time-Series Model Fitting**

The bootstrapped and averaged time series generated by the procedure above were analyzed using the R package paleoTS to compare fit with different evolutionary models. The time-series models compared were those described in paleoTS as URW (unbiased random walk), GRW (directional random walk), Stasis, StrictStasis, and Punc-1 (a single shift). GRW models each consecutive step in the time series by drawing a change in trait value  $(\Delta x)$  from a normal distribution, the mean and variance of which are parameters of the model. URW is the same, but the mean of the  $\Delta x$ distribution is fixed at zero. Stasis draws the new value of x, rather than  $\Delta x$ , from a normal distribution with modeled mean and variance, and StrictStasis is the same but with the variance of the trait distribution fixed at zero. The Punc model is the only complex (changing) model used in this study; it has two segments of Stasis, with different values of mean and variance, plus a single time point at which both are allowed to shift. For precise definitions of the models, see Hunt et al. (2007, 2015) or the paleoTS documentation

(Hunt 2022). The models were compared based on the corrected Akaike information criterion (AICc) weights; the AICc is an estimate of the quality of the model fit that accounts for the number of model parameters, and Akaike weights allow comparison by transforming the AICc values into a vector of proportional support for all tested models. The parameters for the four models were also compared in order to estimate the speed of evolutionary change.

Testing Time Series Length. To capture sensitivity to timeseries length, the model fitting procedure was repeated using two different sets of time bins. As described in the "Paleoclimate Reconstruction" section, the first set has five bins and the other nine. The two sets of time bins are depicted in the vertical lines in Figure 1 and given in Supplemental Table C. For short time series, the number of time points strongly biases the choice of models. The models used vary from one to four parameters, and when the time series has only five points, the AICc comparison strongly favors models with the fewest parameters. On the other hand, dividing an empirically defined time series arbitrarily into smaller intervals tends to make the model selection procedure favor an unbiased random walk (URW). The time-series model fitting results, as summarized in Table 3, reflect this: 90% of the shorter five-step time series were best fit by the simplest model (StrictStasis). This is an unavoidable limitation of the time-series model fitting method.

Omitting the StrictStasis model and fitting only models with two or more parameters transfers the same bias to the two-parameter models (Stasis and URW), but also detects more of the directional trends that appear to the eye. When we split all intervals except the

,		,			
Model name	No. of model parameters	Measurement time series (5 steps)	Measurement time series (9 steps)	PC time series (5 steps)	PC time series (9 steps)
Stasis	2	238	213 (42%)	247	218 (44%)
URW	2	209	214 (41%)	202	166 (33%)
GRW	3	0	68 (13%)	0	108 (22%)
Punc-1,Shift=2	4	66	0	49	0
Punc–1,Shift=3	4	0	18 (4%)	0	6 (1%)

Table 3. Summary statistics for paleoTS time-series model fits on measurement and principal component (PC) time series. GRW, generalized random walk; URW, undirected random walk; Punc-1, punctuated equilibrium with 1 shift. Changes were detected in largely the same set of bones in both five-interval and nine-interval time series, but the shorter time series favored stasis/URW over directional and punctuated over gradual models.

Holocene in half, creating a nine-step time series, this dropped to 70%. Using even smaller time bins starts to create problems with the smaller datasets: if there are too few specimens, the bootstrap resampling procedure creates so much collinearity in the resampled time series that the results are impossible to interpret. To accommodate this effect, we have restricted our time-series model fits to the nine-step time series, because the shorter time series effectively cannot be fit using this method. We have also omitted the Strict-Stasis model and compared only GRW, URW, Stasis, and Punc-1. The results and interpretations given later reflect these decisions.

Testing the Bootstrapping and Averaging Procedure. Comparison of the first 250 bootstrapped time series for each measurement to the averaged time series reveals that the averaging procedure makes the resulting time series less likely to be modeled as stasis. To give a single illustrative example of this effect, although 88% of 250 individual time series of S. fatalis humerus length were best modeled as either URW or Stasis and only 12% as GRW, the GRW model is strongly favored for the averaged time series (Akaike weight 0.58, compared with 0.39 for URW and 0.3 for Stasis) (see Supplementary Fig. D). The same effect holds for the rest of the dataset: the weights of the two-parameter URW and Stasis models are greater for the individual bootstrap replicates than in the averaged time series, while the reverse is true for the threeparameter GRW model (an overall comparison of individual vs. averaged Akaike weights for all nine-step time series is given in Supplementary Fig. E). We interpret this to mean that the averaging procedure successfully filters out some of the noise introduced by the bootstrapping procedure and makes it easier to recover the true signal. We also surmise that the noise introduced by the previous age estimation method, similarly, may have artificially increased the prevalence of stasis as the best-supported model in our original studies.

### Results

The results of the time-series model fits are summarized in Table 3. Three example time series are plotted in Figure 4, as discussed here; to emphasize the diversity of morphological responses among ecologically similar groups in our sample, all three examples are TMTs of extant owl species. Due to the size of the dataset, the complete results are not printed here; instead, all time series are plotted by species in the Supplementary Appendix, and all model weights and model fit parameters for those time series are given in Supplementary Table E.

The majority of nine-step time series were best fit by either a stasis or a random walk (URW) model (83% of measurements and 92% of PCs); and for 23 of the 30 species, all time series are best fit by Stasis and/or URW. As described earlier, this is partially the

effect of a preference for the simplest available model for short time series. However, it is visually apparent from the time-series plots that the size and shape of most of the bones examined here genuinely do not appear to change significantly over the Pleistocene/Holocene transition. The difference between undirected random walk (URW) and stasis (Stasis) is not always evident, and in most cases (29 of 57 bones), the best models for the various measurements and PCs of a single bone are all either Stasis or URW. In a few cases, stasis is clearly a better model for all the time series; for example, Asio otus (Fig. 4A). Much more common is a combination of Stasis and URW, as in Tyto alba (Fig. 4B), in which some time series have significant departures from the mean or even appear to display nonlinear trends; however, detailed interpretation is difficult at this temporal resolution.

The overall evolutionary variance of a trait among time bins is modeled by the "omega" parameter of the Stasis model fit, regardless of the relative weights of the models. Normalizing this parameter by dividing by the mean variance in the last three time intervals, which have relatively narrowly constrained time distributions (see Fig. 2), then taking its square root, we arrive at a value expressed in centimeters that can be used as a measure of the variability of the trait, assuming a static mean through time, which we will refer to as "evolutionary variance." The magnitude of directional trend is reflected in the mstep parameter of the GRW model fit; for ease of discussion, we have divided the time-agnostic values of the model parameters by the average time-step size in order to convert them to millimeters per million years (mm/Myr).

The evolutionary variance of most of our traits is zero: that is, the populations in different time bins are similar enough that no variance is necessary to model them (corresponding to the Strict-Stasis model). This is the case for 51% of the nine-step time series and 65% of the five-step time series, all of which are best modeled as Stasis or URW. Nonzero evolutionary variances are more common in the nine-step time series, which again supports the interpretation that the longer intervals in the five-step series produce larger within-interval variances, impeding our ability to detect change over time.

Directional evolution (GRW) was the favored model for at least one dimension or PC for 28 bones in 13 species. In all three of the bones containing time series for which a stepwise (Punc-1) model was favored, GRW was also favored for other time series. We conclude that the temporal resolution of our data (nine steps) is evidently insufficient to discriminate between stepwise and gradual change. Among the time series best modeled by a directional trend (either GRW or Punc-1), the values of the evolutionary variance parameter are significantly greater than for those for which nondirectional models (URW or Stasis) were favored, and among directional time series, higher evolutionary variance is associated



**Figure 4.** Example measurement and principal component (PC) time series reconstructed from Rancho La Brea (RLB) fossil tarsometatarsi (TMTs) of three extant owls. Colors indicate best-fit time-series model: pink, GRW (directional random walk); red, Punc (a single shift); yellow, URW (unbiased random walk); blue, Stasis. All PCs are plotted together, with wider lines indicating higher PCs, and line colors similarly indicating best fit time-series models. **A**, *Asio otus* TMT: all time series are best fit by a Stasis model, including all PCs. **B**, *Tyto alba* TMT: most time series are best fit by a URW (random walk) model. **C**, *Athene cunicularia* TMT: most time series are best fit by a GRW (directional evolution) model.

with greater weight of the GRW model. Similarly, the magnitudes of the trend parameter in the GRW model (GRW.mstep) are significantly greater among directional time series than among those best fit by the nondirectional models (p = 0.0005). However, the magnitude of the trend parameter was not correlated with the model weight; that is, for a time series displaying a directional evolutionary trend, the size of the trend and its strength were unrelated.

The majority (two-thirds) of the directional trends are toward smaller and/or more gracile forms (Supplementary Table D); trends toward smaller forms also accounted for more variance in the data than those toward larger forms. Figure 4C (Athene cunicularia) is a good example of this tendency, as are some elements in Equus occidentalis and Gymnogyps amplus and calcanea of Camelops hesternus, as discussed later. However, several exceptions to the overall trend are noted. The most notable of these is the sabertoothed cat Smilodon fatalis, in which all the measured postcranial bones increase in size (see Supplementary Appendix A and Supplementary Table D) and are all best fit by directional models with evolutionary rates between 0.1 and 1.3 mm/Myr. This is consistent with other studies on crania in S. fatalis (Meachen et al. 2014; Goswami et al. 2015) but contradicts the study we previously published using the same dataset (Madan et al. 2011). Among the other time series with increasing size, many are either low-variance PCs of otherwise static or shrinking bones (e.g., Aenocyon, Equus, Panthera, Gymnogyps crania) or based on especially small sample numbers (e.g., Lynx), although astragali of Paramylodon and TMTs of a few birds also increase in size and robustness.

### Discussion

### Stasis and Climate Change

*Prevalence of Stasis.* The majority of our results were consistent with those of our previous studies, despite the major methodological difference: the majority of bones in the majority of species remain static in size and shape through the onset and departure of the LGM. Comparing our results with the 9–14% directional evolution benchmark from Hunt et al. (2015) and Voje (2016), we find directional evolution in 13% of nine-step time series of direct measurements and 22% of PCs of variation. Directional evolution in this dataset thus appears to be slightly more common than the average for all morphological time series, although many of the directional changes occurred in noisy low-variance PCs.

There is no indication of larger body size or more robust limb proportions at the LGM, as might be predicted by Bergmann's rule and Allen's rule. If Bergmann's rule was driving local evolution or range shift, we would expect steady or increasing body size and robustness going from the MIS-3 intervals (1–2) into the LGM intervals (3–4), followed by a decrease over the Deglacial and Transitional intervals (5–8). This is not evident in any of the time series. Instead, size and robustness seem to reflect other factors, which may be more easily determined upon individual consideration of each species' particular ecological role. Although a complete evaluation of each species is beyond the scope of this paper, *Gymnogyps californianus* ssp. *amplus* (*G. amplus*) and *Equus occidentalis* both furnish useful examples. Both these species display a robust non-static trend in a large sample size, which allows more detailed interpretation.

All PC 1 time series from these two species are given in Figure 5. In *Equus*, the third metacarpals (forelimb cannon bones) have an obvious linear trend, continually shorter and thinner (-0.2 mm/Myr),



**Figure 5.** Principal component 1 (PC 1) time-series plots for **(A)** all four measured skeletal elements of *Equus occidentalis* and **(B)** forelimbs and hindlimbs of *Gymnogyps amplus (G. californianus)*, with colors as in Fig. 4. Percentage of variation is given in gray at top right of each plot; all PCs have been rotated so that positive = larger. The third metacarpal (cannon bone) of *E. occidentalis* decreases continually along PC 1, but the other three have no directional trend. The three wing bones of *G. amplus* also each decrease along the respective PC 1, but the leg bones have no directional trend. Colors are as in Fig. 4.

interrupted by an upward tick in the first Deglacial interval. This is the only element in *Equus* with such a strong trend: the astragalus becomes somewhat shorter and thicker, the third metatarsal (hindlimb cannon bone) somewhat more gracile, and the patella is static. Notably, the calcanea of *Camelops hesternus* also display a strong trend toward smaller size beginning in the first Deglacial interval (-2.1 mm/Myr).



**Figure 6** Time series of measurements and principal components (PCs) for *Haliaeetus leucocephalus* tarsometatarsi (TMT), with colors as in Fig. 4. The population means are generally static throughout, with two points of apparently ephemeral change standing out: the 18–15 ka population is longer and broader than the mean, and the 11–4 ka point is smaller in all dimensions. Colors are as in Fig. 4.

Although this raises the interesting possibility that these two ecologically similar cohabiting species may have been subject to similar pressures during the Pleistocene/Holocene transition, unfortunately neither calcanea of *E. occidentalis* nor cannon bones of *C. hesternus* were measured, so no further comparison of the two is possible.

The results for *Gymnogyps* are more straightforward to interpret: the two bones with clear downward trends in all dimensions are both from the wing, the carpometacarpus (PC1: -0.2 mm/Myr) and humerus (PC1: -0.2 mm/Myr), and the ulna appears to follow the same trajectory, although the sample is too small to discriminate models. All the lower limb bones (femur, tibiotarsus, and TMT) are static. The skulls have a mixture of different trajectories reflecting their variety of functions; cranial height increases and foramen magnum shrinks, but the majority of skull measurements display no directional change. Evidently wing size in *Gymnogyps* and forelimb length in *Equus* both decreased continually through the transitions into and out of the LGM temperature minimum and thus could not have been responding directly to temperature. These two examples are illustrative of the general lack of apparent temperature correlation among the species studied here, whether static or variable.

Ephemeral Changes. Two particular intervals have anomalous means in enough time series that they merit individual attention. These phenomena are both visible in the time series for Haliaeetus leucocephalus (Fig. 6). First, many species and anatomical elements have a visible difference between the Holocene individuals (11-4 ka) and the rest of the Pleistocene. This shift occurs in the last time bin and is therefore not captured by the time-series analysis. However, this observation is consistent with the conclusions of O'Keefe et al. (2023) regarding the ecological regime shift at the end of the Pleistocene. Interestingly, this appears to apply mainly to extant species. An anomalous interval also often appears somewhere between 20 and 13.3 ka. In Haliaeetus, this appears in the Deglacial B interval (17.5–15 ka), but it occurs one interval earlier or later in some other species, and may be from one to three intervals long. The direction of the change is not consistent. This timing corresponds to the wettest climatic interval in the regional climate record. Further detailed exploration of the validity and possible causes of this potentially interesting phenomenon will not be undertaken in this study.

Methodological Comparison. Assigning all individuals in a pit to the mean age of the pit can, as Fuller et al. (2015: p. 165) put it, "mix samples from very different climate regimes and potentially blur or even obliterate important temporal changes in morphological or isotopic data" by adding a random error to each individual's actual age. This kind of time error would tend to obscure real differences between time bins without introducing any directional bias and therefore would decrease the chance of any statistical time-series method detecting directional evolution. The time-series modeling approach also means that the results returned by this analysis, as well as its predecessors, are biased in favor of low parameter number models, especially for short time series or small samples, which would also disproportionately favor stasis or random walk models. It is encouraging, therefore, that this method detected nonstasis (GRW/Punc) in a number of cases that were not previously noticed. In particular, the new approach is validated by its ability to detect the same trend toward larger size in S. fatalis postcranials that appeared in the mandibles (Meachen et al. 2014) and crania (Goswami et al. 2015), whereas our previous paper using these data failed to detect it (Madan et al. 2011).

A similar comparison can be made for *Aenocyon dirus*: our results here, measured on postcranials, largely echo those of Brannick et al. (2015) on the mandibles using geometric morphometrics. Many of the *A. dirus* measurements in our Deglacial interval are distinctly smaller; this corresponds to their observations of mandibles from Pit 13, which contributes a significant proportion of the individuals in this interval. The mean sizes of postcranials also rebound to their previous averages in the Transitional and Holocene intervals, like their mandibles. We note, though, that the specimens assigned to the Transitional intervals in our study are not *larger* than the overall static average, unlike the Pit 61/67 individuals of Brannick et al. (2015). Perhaps the overlap with Pit 3 in our study has obscured this signal, or perhaps the size increase during the Transitional interval applies only to the mandible.

With this in mind, we must revise our previous conclusions on the following taxa: A. cunicularia, Buteo swainsoni, C. hesternus, Circus hudsonius, Corvus corax, E. occidentalis, Gymnogyps amplus, Paramylodon harlani, and S. fatalis. In these 10 taxa, contra our original publications, the reanalysis indicates that at least some significant directional change did take place in the size/shape of at least one and often several bones. In the other taxa for which some directional change was detected (A. dirus, Caracara prelutosa, Lynx rufus, Panthera atrox, and Sturnella neglecta), the evidence is equivocal, either because the sample size was especially small or because directional trends appeared only in low-variance PCs.

### **Depositional Hiatus**

On the basis of this chronology revision, we can confirm the depositional hiatus noted by Fuller et al. (2014). The interval between 18.5 and 16 uncalibrated <sup>14</sup>C ka BP, which corresponds to approximately 22.4 to 19.3 calendar ka BP, has not been filled by results from subsequent papers. In our analysis, this gap corresponds to the end of the LGM bin and the beginning of the Deglacial bin. The hiatus interval is marked in red on Figure 2, where it can be seen that only three dates fall within it. Of these three, one has unusually large errors, ±3250 years (sample QC405b of Marcus and Berger [1984], Pit 81); one is unreliable due to being from the >30 kDa fraction (sample UCIAMS91653 of Fuller et al. [2014], Pit 61/67); and the third is from a wood sample (sample

427R of Marcus and Berger [1984], Pit 16), which may be unreliable on account of the long environmental residence time of dead wood (Schiffer 1986). For discussion of possible reasons for this hiatus, see Fuller et al. (2014).

### Conclusions

Previous studies of the mammals and birds of RLB by our research group tended to document stasis in most of the morphological variables through most of the late Pleistocene in all 9 mammals and 19 birds that were studied. This was already clear from analysis of variance studies of all the samples (in taxa that were abundant enough for parametric tests), where no pit stood out from the pooled mean of all pits. In particular, there was no evidence of larger individuals with more robust limbs during the peak glacial interval, as would be predicted from Bergmann's rule or Allen's rule, so there was no simple response to climate change in those studies. With newer radiocarbon dates and better estimates of the ages of the specimens, we still do not see an obvious response to the colder conditions of the last glacial interval, but there are indications of a response to the climate change at the glacial/interglacial transition, at the same time that most of the megamammals and many of the birds went extinct. In fact, the patterns are now more complex: stasis or at least no evidence of directional change in the vast majority of the taxa and anatomical elements sampled, but some directional change in a few taxa, none of it correlated in a straightforward fashion with the climate changes in this time interval. Thus, it appears that the birds and mammals of RLB did not respond straightforwardly to climate by becoming smaller as the climate warmed, but some measurements do appear to have responded to the variety of ecological changes during the Pleistocene/Holocene transition.

Several possible follow-up studies are suggested. First, the resampling method in this paper could help to illuminate the effect of temporal uncertainty on time-series modeling results. By generating simulated time series with known evolutionary modes and simulated age uncertainty distributions, resampling them, and assessing the fidelity of the resulting model fits, one could parameterize the recoverability of different evolutionary patterns as a function of time averaging. Second, although we have reevaluated the major claims in the preceding paper series, this review has not fully examined the extensive amount of data presented herein. For 10 of these species there are multiple anatomical elements present in our dataset, which offers an opportunity to examine whether changes in different body parts are coordinated. For example, most of Gymnogyps amplus remains unchanged through the interval, but the cranium, carpometacarpus, and humerus decrease in size. In particular, examining our outlier Smilodon fatalis in detail and in conjunction with other recent studies of dietary isotopes, morphology, and developmental integration in Smilodon (Coltrain et al. 2004; Meachen et al. 2014; Goswami et al. 2015; Fuller et al. 2020) might illuminate the relationship between environment, morphology, and extinction. Comparing the trajectories of species with respect to their shared ecological traits might cast light on the specific environmental pressures affecting the RLB vertebrate community during the Pleistocene/Holocene transition.

**Competing Interests.** The authors declare no competing interests.

**Data Availability Statement.** Supplementary files are available at github.com/vjpsyverson/RLB/size\_glacial/2023Paleobio/Final/Supplementary. All other data and code for this paper are available at github.com/vjpsyverson/

RLB/size\_glacial/2023Paleobio. Time series of measurements will also be uploaded to the Phenotypic Evolution Time Series database (pets.nhm.uio.no/).

Acknowledgments. Our numerous coauthors on the preceding studies in this series, listed in Table 1, collected the data. We thank the reviewers, particularly G. Hunt, for their extensive and insightful comments on the statistical methods, and the editors of the special issue for the invitation to contribute.

### **Literature Cited**

- Ashton, K. G., M. C. Tracy, and A. de Queiroz. 2000. Is Bergmann's Rule valid for mammals? *American Naturalist* 156:390–415.
- Avise, J. C., D. Walker, and G. C. Johns. 1998. Speciation durations and Pleistocene effects on vertebrate phylogeography. *Proceedings of the Royal Society B* 265:1707–1712.
- Balassa, D., D. R. Prothero, and V. J. P. Syverson. 2020. After the saber-tooths: how did cougars and bobcats respond to the end of the Ice Ages? *New Mexico Museum of Natural History Bulletin* 82:1–7.
- Balassa, D., D. R. Prothero, and V. J. P. Syverson. 2022. Stasis in red-tailed hawks (*Buteo jamaicensis*) from La Brea Tar Pits during the last glacial-interglacial cycle. *New Mexico Museum of Natural History Bulletin* **90**.
- Barnosky, A. D. 2005. Effects of Quaternary climatic change on speciation in mammals. *Journal of Mammalian Evolution* 12:247–264.
- Bochenski, Z. M., and K. Campbell. 2006. The extinct California turkey, Meleagris californica, from Rancho La Brea: comparative osteology and systematics. Contributions in Science, Natural History Museum of Los Angeles County 509.
- Brannick, A., J. A. Meachen, and F. R. O'Keefe. 2015. Microevolution of jaw shape in the dire wolf, *Canis dirus*, at Rancho La Brea. Pp. 23–32 in J. Harris, ed. *La Brea and beyond: the paleontology of asphalt-preserved biotas*. Natural History Museum of Los Angeles County Science Series 42. Natural History Museum of Los Angeles County, Los Angeles.
- Bronk Ramsey, C. 2001. Development of the radiocarbon calibration program. *Radiocarbon* 43:355–363.
- Carleton, W. C., and H. S. Groucutt. 2021. Sum things are not what they seem: problems with point-wise interpretations and quantitative analyses of proxies based on aggregated radiocarbon dates. *The Holocene* 31:630–643.
- Cleaveland, C., D. R. Prothero, and V. J. P. Syverson. 2022. How did La Brea Swainson's hawks (*Buteo swainsoni*) respond to climate change? *New Mexico Museum of Natural History Bulletin* 90.
- Coltrain, J. B., J. M. Harris, T. E. Cerling, J. R. Ehleringer, M.-D. Dearing, J. Ward, and J. Allen. 2004. Rancho La Brea stable isotope biogeochemistry and its implications for the palaeoecology of Late Pleistocene coastal southern California. *Palaeogeography, Palaeoclimatology, Palaeoecology* 205:199–219.
- DeAnda, E., D. R. Prothero, and K. Marriott. 2022. How did Grinnell's hawkeagle (Spizaetus grinnelli) from La Brea Tar Pits respond to climate change? New Mexico Museum of Natural History Bulletin 90:155–160.
- **DeSantis, S., and D. R. Prothero**. 2011. Size and shape stasis in late Pleistocene horses and camels from Rancho La Brea during the last glacial-interglacial cycle. *New Mexico Museum of Natural History Bulletin* **53**:505–510.
- Eldredge, N., and S. J. Gould. 1972. Punctuated equilibria: an alternative to phyletic gradualism. Pp. 82–115 in T. J. M. Schopf, ed. *Models in paleobiology*. Freeman, Cooper, San Francisco.
- Estes, S., and S. J. Arnold. 2007. Resolving the paradox of stasis: models with stabilizing selection explain evolutionary divergence on all timescales. *American Naturalist* 169:227–244.
- Falconer, H. 1863. On the American fossil elephant of the regions bordering the Gulf of Mexico (*E. columbi*, Falc.): with general observations on the living and extinct species. *Natural History Review* **10**:43–114.
- Fan, L., T. Cai, Y. Xiong, G. Song, and F. Lei. 2019. Bergmann's rule and Allen's rule in two passerine birds in China. Avian Research 10:34.
- Feakins, S. J., M. S. Wu, C. Ponton, and J. E. Tierney. 2019. Biomarkers reveal abrupt switches in hydroclimate during the last glacial in southern California. Earth and Planetary Science *Letters* 515:164–172.
- Fox, N. S., J. R. Southon, C. M. Howard, G. T. Takeuchi, S. Potze, A. B. Farrell, E. L. Lindsey, and J. L. Blois. 2023. Millennial-scale drivers of small mammal isotopic niche dynamics in southern California. *Palaeogeography, Palaeoclimatology, Palaeoecology* 612:111378.

- Fragomeni, A., and D. R. Prothero. 2011. Stasis in late Quaternary birds from the La Brea tar pits during the last glacial-interglacial cycle. *New Mexico Museum of Natural History Bulletin* 53:511–516.
- Friscia, A. R., B. Van Valkenburgh, L. Spencer, and J. Harris. 2008. Chronology and spatial distribution of large mammal bones in Pit 91, Rancho La Brea. *Palaios* 23:35–42.
- Fuchs, J., J. A. Johnson, and D. P. Mindell. 2012. Molecular systematics of the caracaras and allies (Falconidae: Polyborinae) inferred from mitochondrial and nuclear sequence data. *Ibis* 154:520–532.
- Fuller, B., J. Harris, A. B. Farrell, G. T. Takeuchi, and J. R. Southon. 2015. Sample preparation for radiocarbon dating and isotopic analysis of bone from Rancho La Brea. Pp. 151–167 in J. Harris, ed. La Brea and beyond: the paleontology of asphalt-preserved biotas. Natural History Museum of Los Angeles County Science Series 42. Natural History Museum of Los Angeles County, Los Angeles.
- Fuller, B. T., S. M. Fahrni, J. M. Harris, A. B. Farrell, J. B. Coltrain, L. M. Gerhart, J. K. Ward, R. E. Taylor, and J. R. Southon. 2014. Ultrafiltration for asphalt removal from bone collagen for radiocarbon dating and isotopic analysis of Pleistocene fauna at the tar pits of Rancho La Brea, Los Angeles, California. *Quaternary Geochronology* 22:85–98.
- Fuller, B. T., J. R. Southon, S. M. Fahrni, A. B. Farrell, G. T. Takeuchi, O. Nehlich, E. J. Guiry, M. P. Richards, E. L. Lindsey, and J. M. Harris. 2020. Pleistocene paleoecology and feeding behavior of terrestrial vertebrates recorded in a pre-LGM asphaltic deposit at Rancho La Brea, California. *Palaeogeography, Palaeoclimatology, Palaeoecology* 537:109383.
- Futuyma, D. J. 2010. Evolutionary constraint and ecological consequences. Evolution 64:1865–1884.
- Gill, B. J. 2003. Osteometry and systematics of the extinct New Zealand ravens (Aves: Corvidae: *Corvus*). *Journal of Systematic* Palaeontology 1:43–58.
- Gillespy, P., D. R. Prothero, and V. J. P. Syverson. 2016. Stasis in teratorns from the La Brea tar pits during the last glacial-interglacial cycle. *New Mexico Museum of Natural History Bulletin* 74:67–71.
- Goswami, A., W. J. Binder, J. Meachen, and F. R. O'Keefe. 2015. The fossil record of phenotypic integration and modularity: a deep-time perspective on developmental and evolutionary dynamics. *Proceedings of the National Academy of Sciences USA* 112:4891–4896.
- Gould, S. J. 2002. The structure of evolutionary theory. Belknap Press, Cambridge, Mass.
- Gould, S. J., and N. Eldredge. 1993. Punctuated equilibrium comes of age. Nature 366:223–227.
- Graham, R. W., E. L. Lundelius, M. A. Graham, E. K. Schroeder, R. S. Toomey, E. Anderson, A. D. Barnosky, et al. 1996. Spatial response of mammals to late Quaternary environmental fluctuations. *Science* 272:1601–1606.
- Heusser, L. E. 1995. Pollen stratigraphy and paleoecologic interpretation of the 160-k.y. record from Santa Barbara Basin, Hole 893A. In J. P. Kennett, J. G. Baldauf, and M. Lyle, eds. Proceedings of the Ocean Drilling Program, Scientific Results 146:265–279. College Station, Tex.
- Heusser, L. E. 1998. Direct correlation of millennial-scale changes in western North American vegetation and climate with changes in the California Current System over the past ~60 kyr. *Paleoceanography* 13:252–262.
- Heusser, L. E., and F. Sirocko. 1997. Millennial pulsing of environmental change in southern California from the past 24 k.y.: a record of Indo-Pacific ENSO events? *Geology* 25:243–246.
- Heusser, L. E., M. E. Kirby, and J. E. Nichols. 2015. Pollen-based evidence of extreme drought during the last glacial (32.6–9.0 ka) in coastal southern California. *Quaternary Science Reviews* 126:242–253.
- Hinz, M., C. Schmid, D. Knitter, and C. Tietze. 2021. oxcAAR: interface to "OxCal" radiocarbon calibration, R package version 1.1.1. https://CRAN.Rproject.org/package=oxcAAR.
- Holden, A. R., and J. R. Southon. 2016. Radiocarbon dating and stable isotopic analysis of insect chitin from the Rancho La Brea Tar Pits, Southern California. *Radiocarbon* 58:99–113.
- Holden, A. R., J. R. Southon, K. Will, M. E. Kirby, R. L. Aalbu, and M. J. Markey. 2017. A 50,000 year insect record from Rancho La Brea, Southern California: insights into past climate and fossil deposition. *Quaternary Science Reviews* 168:123–136.
- Hunt, G. 2007. The relative importance of directional change, random walks, and stasis in the evolution of fossil lineages. *Proceedings of the National Academy of Sciences USA* 104:18404–18408.

- Hunt, G. 2022. paleoTS: analyze paleontological time-series, R package version 0.5.3. https://CRAN.R-project.org/package=paleoTS.
- Hunt, G., M. J. Hopkins, and S. Lidgard. 2015. Simple versus complex models of trait evolution and stasis as a response to environmental change. *Proceedings of the National Academy of Sciences USA* 112:4885–4890.
- Jablonski, D. 2008. Species selection: theory and data. Annual Review of Ecology, Evolution, and Systematics 39:501–524.
- Janis, C.M. 1989. A climatic explanation for patterns of evolutionary diversity in ungulate mammals. *Palaeontology* 32:463–481.
- Janis, C. M. 1993. Tertiary mammal evolution in the context of changing climates, vegetation, and tectonic events. *Annual Reviews of Ecology and Systematics* 24:467–500.
- Johnsgard, P. A. 1990. Hawks, eagles, & falcons of North America: biology and natural history. *Paul Johnsgard Collection* **52**. https://digitalcommons.un l.edu/johnsgard/52
- Kirby, M. E., S. J. Feakins, N. Bonuso, J. M. Fantozzi, and C. A. Hiner. 2013. Latest Pleistocene to Holocene hydroclimates from Lake Elsinore, California. *Quaternary Science Reviews* 76:1–15.
- Kirby, M. E., L. Heusser, C. Scholz, R. Ramezan, M. A. Anderson, B. Markle, E. Rhodes, et al. 2018. A late Wisconsin (32–10k cal a BP) history of pluvials, droughts and vegetation in the Pacific south-west United States (Lake Elsinore, CA). *Journal of Quaternary Science* 33: 238–254.
- Kirby, M. E. C., Patterson, W. P., Lachniet, M., Noblet, J. A., Anderson, M. A., Nichols, K., & Avila, J. (2019). Pacific Southwest United States Holocene Droughts and Pluvials Inferred From Sediment δ18O(calcite) and Grain Size Data (Lake Elsinore, California). Frontiers in Earth Science 7. https://doi.org/ 10.3389/feart.2019.00074
- Long, K. L., D. R. Prothero, and V. J. P. Syverson. 2016. Stasis in Rancho La Brea black vultures (*Coragyps occidentalis*) over the last glacial-interglacial cycle. *New Mexico Museum of Natural History Bulletin* 74:121–125.
- Long, K. L., D. R. Prothero, and V. J. P. Syverson. 2020. How do small birds evolve in response to climate change? Data from the long-term record at La Brea Tar Pits. *Integrative Zoology* 15:249–261.
- Madan, M., D. R. Prothero, and A. Sutyagina. 2011. Did Rancho La Brea large felids (*Panthera atrox* and *Smilodon fatalis*) change in size or shape during the Late Pleistocene? *New Mexico Museum of Natural History Bulletin* 53:554–563.
- Madan, M. A., D. R. Prothero, and V. J. P. Syverson. 2015. Stasis in great horned owls from the La Brea tar pits during the last glacial-interglacial cycle. *New Mexico Museum of Natural History Bulletin* 65:221–225.
- Madan, M. A., D. R. Prothero, and V. J. P. Syverson. 2016. Stasis in barn owls from the La Brea tar pits during the last glacial-interglacial cycle. *New Mexico Museum of Natural History Bulletin* 74:153–157.
- Madan, M. A., D. R. Prothero, and V. J. P. Syverson. 2019. Stasis in the smaller owls from Rancho La Brea during the last glacial-interglacial climate change. *Paleontologica Electronica* 22:3:70.
- Marcus, L. F., and R. Berger. 1984. The significance of radiocarbon dates for Rancho La Brea. Pp. 159–183 *in* P. S. Martin and R. G. Klein, eds. *Quaternary extinctions: a prehistoric revolution*. University of Arizona Press, Tucson.
- Marriott, K., D. R. Prothero, and K. I. Watmore. 2022. How did northern harriers (*Circus hudsonius*) from La Brea Tar Pits respond to climate change during the last glacial-interglacial cycle? *New Mexico Museum of Natural History Bulletin* 90:303–308.
- Mayr, E. 1956. Geographical character gradients and climatic adaptation. *Evolution* 10:105–108.
- McGill, B. J., E. A. Hadly, and B. A. Maurer. 2005. Community inertia of Quaternary small mammal assemblages in North America. *Proceedings of the National Academy of Sciences USA* 102:16701–16706.
- Meachen, J. A., and J. X. Samuels. 2012. Evolution in coyotes (*Canis latrans*) in response to the megafaunal extinctions. *Proceedings of the National Academy* of Sciences USA 109:4191–4196.
- Meachen, J. A., F. R. O'Keefe, and R. W. Sadler. 2014. Evolution in the sabretooth cat, *Smilodon fatalis*, in response to Pleistocene climate change. *Journal* of Evolutionary Biology 27:714–723.
- Molina, S., and D. R. Prothero. 2011. Stasis in late Quaternary golden eagles from La Brea Tar Pits over the last glacial-interglacial cycle. *New Mexico Museum of Natural History Bulletin* 53:564–569.

- Nowak, R. M. 1999. Walker's mammals of the world. Johns Hopkins University Press, Baltimore, Md.
- O'Keefe, F. R., E. V. Fet, and J. M. Harris. 2009. Compilation, calibration, and synthesis of faunal and floral radiocarbon dates, Rancho La Brea, California. *Contributions in Science, Natural History Museum of Los Angeles County* **518**: 1–16.
- O'Keefe, F. R., R. E. Dunn, E. M. Weitzel, M. R. Waters, L. N. Martinez, W. J. Binder, J. R. Southon, et al. 2023. Pre–Younger Dryas megafaunal extirpation at Rancho La Brea linked to fire-driven state shift. *Science* 381: 3594.
- Olson, S., D. R. Prothero, and V. J. P. Syverson. 2022. Stasis in Neophrontops americanus (Egyptian vultures) from La Brea Tar Pits during the last glacialinterglacial cycle. New Mexico Museum of Natural History Bulletin 90:325– 330.
- Oswald, J. A., J. M. Allen, K. E. Witt, R. A. Folk, N. A. Albury, D. W. Steadman, and R. P. Guralnick. 2019. Ancient DNA from a 2,500-year-old Caribbean fossil places an extinct bird (*Caracara creightoni*) in a phylogenetic context. *Molecular Phylogenetics and Evolution* 140:106576.
- Peterson, A. T. 2011. Ecological niche conservatism: a time-structured review of evidence. *Journal of Biogeography* 38:817–827.
- Peterson, A. T., J. Soberón, and V. Sánchez-Cordero. 1999. Conservatism of ecological niches in evolutionary time. *Science* 285:1265–1267.
- Prost, S., J. Klietmann, T. van Kolfschoten, R. P. Guralnick, E. Waltari, K. Vrieling, M. Stiller, et al. 2013. Effects of Late Quaternary climate change on Palearctic shrews. *Global Change Biology* 19:1865–1874.
- Prothero, D.R. 1999. Does climatic change drive mammalian evolution? GSA Today 9:1–5.
- Prothero, D. R., and K. R. Raymond. 2011. Stasis in the Late Pleistocene ground sloths (*Paramylodon harlani*) from Rancho La Brea Tar Pits, California. New Mexico Museum of Natural History Bulletin 53:624–628.
- Prothero, D. R., V. Syverson, K. R. Raymond, M. Madan, S. Molina, A. Fragomeni, S. DeSantis, A. Sutyagina, and G. L. Gage. 2012. Size and shape stasis in Late Pleistocene mammals and birds from Rancho La Brea during the Last Glacial–Interglacial cycle. *Quaternary Science Reviews* 56:1–10.
- Pyron, R. A., G. C. Costa, M. A. Patten, and F. T. Burbrink. 2015. Phylogenetic niche conservatism and the evolutionary basis of ecological speciation. *Biological Reviews* 90:1248–1262.
- Raymond, K. R., and D. R. Prothero. 2011. Did climate change affect size in Late Pleistocene bison? New Mexico Museum of Natural History Bulletin 53: 636–640.
- Reimer, P. J., W. E. N. Austin, E. Bard, A. Bayliss, P. G. Blackwell, C. Bronk Ramsey, M. Butzin, et al. 2020. The IntCal20 Northern Hemisphere radiocarbon age calibration curve (0–55 cal kBP). *Radiocarbon* 62:725–757.
- Rodríguez, M. Á., M. Á. Olalla-Tárraga, and B. A. Hawkins. 2008. Bergmann's rule and the geography of mammal body size in the Western Hemisphere. *Global Ecology and Biogeography* 17:274–283.
- Santos, S. M., D. R. Prothero, and K. Marriott. 2022. Stasis in the extinct La Brea fragile eagle (*Buteogallus fragilis*) in response to climate change. New Mexico Museum of Natural History Bulletin 90:365–370.
- Schiffer, M. B. 1986. Radiocarbon dating and the "old wood" problem: the case of the Hohokam chronology. *Journal of Archaeological Science* 13(1):13–30.
- Secord, R., J. I. Bloch, S. G. B. Chester, D. M. Boyer, A. R. Wood, S. L. Wing, M. J. Kraus, F. A. McInerney, and J. Krigbaum. 2012. Evolution of the earliest horses driven by climate change in the Paleocene–Eocene Thermal Maximum. Science 335:959–962.
- Smith, F. A., J. L. Betancourt, and J. H. Brown. 1995. Evolution of body size in the woodrat over the past 25,000 years of climate change. *Science* 270:2012–2014.
- Stanley, S. M. 1975. A theory of evolution above the species level. Proceedings of the National Academy of Sciences USA 72:646–650.
- Stucky, R.K. 1992. Mammalian faunas in North America of Bridgerian to Early Arikareean "Ages" (Eocene and Oligocene). *Eocene–Oligocene Climatic and Biotic Evolution*, eds Prothero, DR, Berggren, WA, (Princeton Univ Press, Princeton), pp 464–493.
- Sunquist, M., and F. Sunquist. 2002. Wild cats of the world. University of Chicago Press, Chicago.
- Syverson, V., and D. R. Prothero. 2010. Evolutionary patterns in late Quaternary California Condors. PalArch Journal of Vertebrate Paleontology 7:1–18.

- Uyeda, J. C., T. F. Hansen, S. J. Arnold, and J. Pienaar. 2011. The million-year wait for macroevolutionary bursts. *Proceedings of the National Academy of Sciences USA* 108:15908–15913.
- Voje, K. L. 2016. Tempo does not correlate with mode in the fossil record. Evolution 70:2678–89.
- Vrba, E. S., and S. J. Gould. 1986. The hierarchical expansion of sorting and selection: sorting and selection cannot be equated. *Paleobiology* 12:217–228.
- Ward, J. K., J. M. Harris, T. E. Cerling, A. Wiedenhoeft, M. J. Lott, M.-D. Dearing, J. B. Coltrain, and J. R. Ehleringer. 2005. Carbon starvation in glacial trees recovered from the La Brea Tar Pits, southern California. *Proceedings of the National Academy of Sciences USA* 102:690–694.
- Watmore, K. I., and D. R. Prothero. 2022. The effects of climate change on ferruginous hawks (*Buteo regalis*) from La Brea Tar Pits during the last glacialinterglacial cycle. *New Mexico Museum of Natural History Bulletin* 89:455–460.
- Wiens, J. J. 2004. Speciation and ecology revisited: phylogenetic niche conservatism and the origin of species. *Evolution* 58:193–197.

- Willis, K. J., K. D. Bennett, D. Walker, K. J. Willis, and K. J. Niklas. 2004. The role of Quaternary environmental change in plant macroevolution: the exception or the rule? *Philosophical Transactions of the Royal Society B* 359:159–172.
- Woodard, G. D., and L. F. Marcus. 1973. Rancho La Brea fossil deposits: a re-evaluation from stratigraphic and geological evidence. *Journal of Paleontology* 47:54–69.
- Yom-Tov, Y., and E. Geffen. 2011. Recent spatial and temporal changes in body size of terrestrial vertebrates: probable causes and pitfalls. *Biological Reviews* 86:531–541.
- Zink, R. M., J. Klicka, and B. R. Barber. 2004. The tempo of avian diversification during the Quaternary. *Philosophical Transactions of the Royal Society B* 359:215–220.
- Zink, R. M., S. Botero-Cañola, H. Martinez, and K. M. Herzberg. 2020. Niche modeling reveals life history shifts in birds at La Brea over the last twenty millennia. *PLoS ONE* 15:e0227361.